Author's response to reviews

Title: Tooth-size discrepancy and arch form: is there a link?

Authors:

Abdullah M Aldrees (amaldrees@ksu.edu.sa)
Abdulmajeed M Al-Shujaa (o_ams_o@hotmail.com)
Mohammad A Alqahtani (malqahta@buffalo.edu)
Ali S Aljhani (aljhani.a@gmail.com)

Version: 2
Date: 8 November 2014

Author's response to reviews:

The requested changes were made in the revised manuscript. The following sections were added: List of Abbreviations, Availability of Supporting Data, Competing Interests, Authors Contributions. In addition, the line numbers were made continuous.